Author's response to reviews

Title: Pattern of Traditional Medicine Use by Adult Saudi Patients with Neurological Disorders

Authors:

Yousef M Mohammad (ymohammad@ksu.edu.sa)
Ahmed N Al-Ahmari (An1410@gmail.com)
Fahad A Al-Dashash (F_a_dashash@gmail.com)
Fawaz A Al-Hussain (Faalhussain@ksu.edu.sa)
Firas F Al-Mansour (Firas_dm@hotmail.com)
Abdullah A Masoud (A_stonecold_a@hotmail.com)
Hoda a Jradi (jradiho@NGHA.MED.SA)

Version: 7 Date: 4 January 2015

Author's response to reviews: see over
Editorial Comments:

We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

The manuscript was edited by Philip Embleton, who is a writer editor I. The changes he made on the manuscript are highlighted in red.